Author's response to reviews

**Title:** Low intensity blue-enriched white light (750 lux) and standard bright light (10 000 lux) are equally effective in treating SAD

**Authors:**

Ybe Meesters (y.meesters@psy.umcg.nl)
Vera Dekker (veradekker@gmail.com)
Luc J.M. Schlangen (luc.schlangen@philips.com)
Elske H. Bos (e.h.bos01@med.umcg.nl)
Martine J. Ruiter (m.j.ruiter@psy.umcg.nl)

**Version:** 2  **Date:** 16 August 2010

**Author's response to reviews:** see over
Dear Sir,

Enclosed please find the manuscript of the article

Low-intensity blue-enriched white light (750 lux) and standard bright light (10 000 lux) are equally effective in treating SAD

by Y. Meesters, V. Dekker’, L.J.M. Schlangen, E.H. Bos, and M.J. Ruiter

for publication in BMC Psychiatry.

We have rewritten the manuscript in the journal style and hope that the process can continue now.

We do hope that our article is acceptable for publication in BMC psychiatry

Awaiting your decision,

Yours sincerely,

Ybe Meesters, PhD

University Medical Center Groningen
University Center for Psychiatry
Email: y.meesters@sy.umcg.nl